Author's response to reviews

Title: Adapting the Adult Social Care Outcomes Toolkit (ASCOT) for use in care home quality monitoring: conceptual development and testing.

Authors:

Ann-Marie Towers (A.Towers@kent.ac.uk)
Jacquetta M Holder (J.M.Holder@kent.ac.uk)
Nick J Smith (N.J.Smith@kent.ac.uk)
Tanya Crowther (T.Crowther@kent.ac.uk)
Ann P Netten (A.P.Netten@kent.ac.uk)
Elizabeth Welch (E.Welch@kent.ac.uk)
Grace Collins (G.Collins@kent.ac.uk)

Version: 3 Date: 25 June 2015

Author's response to reviews:

Dear Editors,

Please find attached the revised version of the article, with changes made in response to the editors' comments.

The following changes have been made:
- Specific statement of informed consent has been added to the end of the methods section (lines 206-209).
- Grace Collins has been changed to GC (see line 553)
- Juliette Malley has been added to the acknowledgements
- The heading of the first column in Table 3 has been corrected to CH4-HL to match the rest of the article.

Yours Faithfully,

Ann-Marie Towers